Dear Editor,

I have made the following changes to the article:-

1. Title and Abstract are identical in the manuscript and abstract.

2. The author list has been upgraded. In particular I have changed the authors to meet the criteria outlined: first name, then initial for second name and then surname for each author where this was appropriate.

3. The title has been changed to sentence case.

4. Running title has been removed.

5. Keywords section has been removed.

6. Tables have been checked. We have reformatted the tables so there are no merged cells (all tables are in Simple 2 format)

7. The figures have been changed from 1a.,1b etc to 1,2,3. I have adjusted the text (results section) and the figure legend to read accordingly.

8. The figures have been saved as high quality jpegs.